Author’s response to reviews

Title: Seasonal influenza vaccination coverage and its determinants among nursing homes personnel in Western France.

Authors:

Christelle ELIAS (christelle137@hotmail.com)
Anna FOURNIER (annafournier87@gmail.com)
Anca VASILIU (ancavasiliuro@yahoo.com)
Nicolas BEIX (nicobeix@gmail.com)
Rémi DEMILLAC (demillac@club-internet.fr)
Hélène TILLAUT (helene.tillaut@ars.sante.fr)
Yvonnick GUILLOIS (Yvonnick.GUILLOIS@ars.sante.fr)
Serge EYEBE (drsergeeyebe@netcourrier.com)
Bastien MOLLO (bastien.mollo@gmail.com)
Pascal CREPEY (crepey@gmail.com)

Version: 3 Date: 28 Jun 2017

Author’s response to reviews:

Following editorial office request, we have added the following explanation within the 'Ethics and consent to participate' section of your manuscript: "We consulted the legal and ethic department of Santé Publique France (the French National Agency for Public Health), who stated that the study falls under the definition of "non interventional" because it was "based exclusively on administering a questionnaire without any change in person care or in monitoring procedure". Hence, they state that this kind of "observational" study did not require the approval of an ethic committee (CPP)."